Author's response to reviews

Title: Gender differences in the use of transportation services to community rehabilitation programs

Authors:

Nanako Tamiya (ntamiya@md.tsukuba.ac.jp)
Li-Mei Chen (chen@kwansei.ac.jp)
Yasuki Kobayashi (yasukiik@m.u-tokyo.ac.jp)
Mariko Kaneda (mariko.kaneda@city.arakawa.tokyo.ac.jp)
Eiji Yano (eyano@med.teikyo-u.ac.jp)

Version: 8 Date: 2 June 2009

Author's response to reviews:

May 19, 2009

Editor-in-Chief, Melissa Norton, MD

Dear Dr. Norton:

On behalf of my co-authors, thank you very much for your core pre-acceptance of our manuscript, “Gender differences in the use of transportation services to community rehabilitation programs,” to your journal, BMC Geriatrics.

We have made the changes advised in your email, please view the manuscript via internet submission system. If there are any other changes you’d like us to make, please do not hesitate to let us know.

We would like to know if it is possible for us to receive a letter of acceptance at your earliest convenience? It would be very much appreciated if the letter could be sent via email before May 27.

Thank you very much for allowing us to share our manuscript with your global readership.

Sincerely yours,

Li-Mei Chen, PhD
Kwansei Gakuin University